

Safety and immunogenicity of 2-dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination in children and adolescents in Africa: A randomised, placebo-controlled, multicentre Phase II clinical trial

Zacchaeus Anywaine, Houreratou Barry, Omu Anzala, Gaudensia Mutua, Sodiomon Sirima, Serge Eholie, Hannah Kibuuka, Christine Bétard, Laura Richert, Christine Lacabaratz, et al.

▶ To cite this version:

Zacchaeus Anywaine, Houreratou Barry, Omu Anzala, Gaudensia Mutua, Sodiomon Sirima, et al.. Safety and immunogenicity of 2-dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination in children and adolescents in Africa: A randomised, placebo-controlled, multicentre Phase II clinical trial. PLoS Medicine, 2022, 19 (1), pp.e1003865. 10.1371/journal.pmed.1003865 . hal-03690128

HAL Id: hal-03690128 https://inria.hal.science/hal-03690128v1

Submitted on 11 Dec 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Anywaine Z, Barry H, Anzala O, Mutua G, Sirima SB, Eholie S, et al. (2022) Safety and immunogenicity of 2-dose heterologous Ad26. ZEBOV, MVA-BN-Filo Ebola vaccination in children and adolescents in Africa: A randomised, placebocontrolled, multicentre Phase II clinical trial. PLoS Med 19(1): e1003865. https://doi.org/10.1371/ journal.pmed.1003865

Academic Editor: James G. Beeson, Burnet Institute, AUSTRALIA

Received: May 20, 2021

Accepted: November 9, 2021

Published: January 11, 2022

Copyright: © 2022 Anywaine et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Janssen has an agreement with the Yale Open Data Access (YODA) Project to serve as the independent review panel for evaluation of requests for clinical study reports and participant level data from investigators and physicians for scientific research that will advance medical knowledge and public health. Data will be made available following publication and approval by YODA of any formal requests with a defined analysis plan. For more information on this process RESEARCH ARTICLE

Safety and immunogenicity of 2-dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination in children and adolescents in Africa: A randomised, placebo-controlled, multicentre Phase II clinical trial

Zacchaeus Anywaine¹, Houreratou Barry², Omu Anzala³, Gaudensia Mutua³, Sodiomon B. Sirima⁴, Serge Eholie⁵, Hannah Kibuuka⁶, Christine Bétard⁷, Laura Richert^{7,8}, Christine Lacabaratz^{8,9}, M. Juliana McElrath¹⁰, Stephen C. De Rosa¹⁰, Kristen W. Cohen¹⁰, Georgi Shukarev¹¹, Michael Katwere¹¹, Cynthia Robinson¹¹, Auguste Gaddah¹², Dirk Heerwegh¹², Viki Bockstal¹¹, Kerstin Luhn¹¹, Maarten Leyssen¹¹, Rodolphe Thiébaut^{7,8}, Macaya Douoguih¹¹, on behalf of the EBL2002 Study group¹¹

 Medical Research Council/Uganda Virus Research Institute and London School of Hygiene and Tropical Medicine Uganda Research Unit, Entebbe, Uganda, 2 Centre MURAZ, Bobo-Dioulasso, Burkina Faso,
KAVI - Institute of Clinical Research University of Nairobi, Nairobi, Kenya, 4 Centre National de Recherche et de Formation sur le Paludisme (CNRFP), Unité de Recherche Clinique de Banfora, Banfora, Burkina Faso,
Unit of Infectious and Tropical Diseases, BPV3, Treichville University Teaching Hospital, Abidjan, Côte d'Ivoire, 6 Makerere University - Walter Reed Project, Kampala, Uganda, 7 Univ. Bordeaux, Inserm, Bordeaux Population Health Research Center, UMR 1219; Inria SISTM team; CHU Bordeaux; CIC 1401, EUCLID/F-CRIN Clinical Trials Platform, Bordeaux, France, 8 Vaccine Research Institute (VRI), Créteil, France, 9 Université Paris-Est Créteil, Faculté de Médecine, INSERM U955, Team 16, Créteil, France, 10 Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, Washington, United States of America, 11 Janssen Vaccines and Prevention, Leiden, the Netherlands, 12 Janssen Research & Development, Beerse, Belgium

¶ Membership of the EBL2002 Study group is provided in the Acknowledgements * rodolphe.thiebaut@u-bordeaux.fr

Abstract

Background

Reoccurring Ebola outbreaks in West and Central Africa have led to serious illness and death in thousands of adults and children. The objective of this study was to assess safety, tolerability, and immunogenicity of the heterologous 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccination regimen in adolescents and children in Africa.

Methods and findings

In this multicentre, randomised, observer-blind, placebo-controlled Phase II study, 131 adolescents (12 to 17 years old) and 132 children (4 to 11 years old) were enrolled from Eastern and Western Africa and randomised 5:1 to receive study vaccines or placebo. Vaccine groups received intramuscular injections of Ad26.ZEBOV (5×10^{10} viral particles) and MVA-BN-Filo (1×10^{8} infectious units) 28 or 56 days apart; placebo recipients received saline. or to make a request, please visit The Yoda Project site at http://yoda.yale.edu. The data-sharing policy of Janssen Pharmaceutical Companies of Johnson & Johnson is available at https://www.janssen. com/clinical-trials/transparency.

Funding: This work has been supported by Janssen Vaccines and Prevention B.V., the European Commission IMI2 programme under EBOVAC2 (GA 115861), part of the Innovative Medicines Initiative Ebola+ Program and the National Institute for Health Research (NIHR) Oxford Biomedical Research Centre. Funding for medical writing assistance and publication coordination was provided by Janssen Vaccines and Prevention B.V. Eight authors (GS, CR, AG, DH. VB. KL. ML. and MD) were full-time employees of the study sponsor (Janssen Vaccines and Prevention) and were involved in study design, data collection and analysis, data interpretation, and preparation of the manuscript. The sponsor performed data management and data analyses.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: ZA, HB, CB, LR, CL and RT report grants from IMI2-2 [Grant Agreement EBOVAC2 (No.115861) from the Innovative Medicines Initiative 2 Joint Undertaking which receives support from the European Union's Horizon 2020 research and innovation programme] during the conduct of the study. SBS reports grants from Janssen Vaccines & Prevention B.V. during the conduct of the study. MK was a full-time employee of Janssen, Pharmaceutical Companies of Johnson & Johnson at the time of the study. GS, CR, AG, DH, VB, KL, ML and MD were full-time employees of Janssen, Pharmaceutical Companies of Johnson & Johnson at the time of the study, and may own shares in Janssen, Pharmaceutical Companies of Johnson & Johnson. All other authors have nothing to disclose.

Abbreviations: AE, adverse event; CI, confidence interval; EBOV, Ebola virus; ELISpot, enzyme-linked immunospot; EU, ELISA units; EVD, Ebola virus disease; FANG, Filovirus Animal Nonclinical Group; GMC, geometric mean concentration; GMT, geometric mean titre; GP, glycoprotein; IC₅₀, half maximal inhibitory titre; IC₉₀, 90% inhibitory concentration; ICS, intracellular cytokine staining; IFN, interferon; IL, interleukin; Inf.U, infectious units; IWRS, interactive web response system; LLOQ, lower limit of quantification; PBMC, peripheral blood mononuclear cell; psVNA, pseudovirion neutralisation assay; SAE, serious adverse event; SFU, spot-forming units; TNF, tumour necrosis factor. Primary outcomes were safety and tolerability. Solicited adverse events (AEs) were recorded until 7 days after each vaccination and serious AEs (SAEs) throughout the study. Secondary and exploratory outcomes were humoral immune responses (binding and neutralising Ebola virus [EBOV] glycoprotein [GP]-specific antibodies), up to 1 year after the first dose. Enrolment began on February 26, 2016, and the date of last participant last visit was November 28, 2018. Of the 263 participants enrolled, 217 (109 adolescents, 108 children) received the 2-dose regimen, and 43 (20 adolescents, 23 children) received 2 placebo doses. Median age was 14.0 (range 11 to 17) and 7.0 (range 4 to 11) years for adolescents and children, respectively. Fifty-four percent of the adolescents and 51% of the children were male. All participants were Africans, and, although there was a slight male preponderance overall, the groups were well balanced. No vaccine-related SAEs were reported; solicited AEs were mostly mild/moderate. Twenty-one days post-MVA-BN-Filo vaccination, binding antibody responses against EBOV GP were observed in 100% of vaccinees (106 adolescents, 104 children). Geometric mean concentrations tended to be higher after the 56-day interval (adolescents 13,532 ELISA units [EU]/mL, children 17,388 EU/mL) than the 28-day interval (adolescents 6,993 EU/mL, children 8,007 EU/mL). Humoral responses persisted at least up to Day 365.

A limitation of the study is that the follow-up period was limited to 365 days for the majority of the participants, and so it was not possible to determine whether immune responses persisted beyond this time period. Additionally, formal statistical comparisons were not preplanned but were only performed post hoc.

Conclusions

The heterologous 2-dose vaccination was well tolerated in African adolescents and children with no vaccine-related SAEs. All vaccinees displayed anti-EBOV GP antibodies after the 2-dose regimen, with higher responses in the 56-day interval groups. The frequency of pyrexia after vaccine or placebo was higher in children than in adolescents. These data supported the prophylactic indication against EBOV disease in a paediatric population, as licenced in the EU.

Trial registration

ClinicalTrials.gov NCT02564523.

Author summary

Why was the study done?

- There have been larger and more extensive Ebola virus disease (EVD) outbreaks in Africa in the past decade with no licenced treatments available. As such, there is an unmet medical need for prophylactic Ebola vaccines.
- This study was performed to evaluate whether a 2-dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination was safe and immunogenic in healthy African children.

What did the researchers do and find?

- In this randomised, placebo-controlled, Phase II clinical trial, the Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination regimen was administered to African participants in 2 age cohorts (12 to 17 and 4 to 11 years).
- No vaccine-related serious adverse events were reported, and robust immune responses were induced in both adolescents and children after receiving the active 2-dose regimen.

What do these findings mean?

- These data support the use of the Ad26.ZEBOV, MVA-BN-Filo vaccination regimen in African adolescents and children at risk of Ebola infection.
- Although vaccination according to a 28-day regimen may lead to protection against EVD in a shorter time frame, vaccination according to a 56-day regimen results in higher EBOV GP binding and neutralising antibody responses.
- The observation that Ad26 preexisting immunity in the majority of participants does not affect the EBOV GP-specific antibody responses postvaccination augurs well for the use of this vaccine regimen even in regions with a high prevalence of preexisting Ad26 seropositivity.

Introduction

Ebola disease due to the Ebola virus (EBOV) has been responsible for several major outbreaks in Africa since first being identified in 1976 [1]. The 2 largest outbreaks have been in Guinea, Liberia, and Sierra Leone (2014 to 2016) [2], and the Democratic Republic of the Congo (2018 to 2020) [3]. The lack of effective therapy and the lethality of Ebola virus disease (EVD) makes effective vaccination a major medical need, which has driven several vaccine development programmes, typically based on the presentation of the viral surface glycoprotein (GP).

A 1-dose, recombinant, replication-competent vesicular stomatitis viral vectored vaccine expressing the Kikwit GP (rVSV-ZEBOV-GP, Ervebo, Merck) demonstrated 97.5% to 100% efficacy when used in a ring-vaccination strategy, and it was approved by the FDA and EMA for use in persons \geq 18 years [4–7]. Janssen Vaccines & Prevention B.V. has developed a 2-dose heterologous regimen, which recently received approval under exceptional circumstances by the EMA for prophylactic use in persons aged 1 year and older [8–10]. This regimen comprises Ad26.ZEBOV (Zabdeno) and MVA-BN-Filo (Mvabea) administered approximately 8 weeks apart. Phase I/II studies in European and African adults established acceptable safety, tolerability, and robust immunogenicity of the 2-dose regimen with intervals of 28 or 56 days between vaccinations [11–16]. Both vaccines are recommended by SAGE in outbreak settings for infants and children from birth to 17 years of age [17].

We previously reported a Phase II study in healthy and HIV-infected adult African participants to assess the safety and immunogenicity of different timing intervals between vaccinations in a heterologous regimen [16]. This report presents the safety and immunogenicity results in 12- to 17-year-old adolescents and 4- to 11-year-old children from the same study.

Methods

Study overview

This randomised, observer-blind, placebo-controlled Phase II study was performed in 7 sites in 4 African countries: Burkina Faso (Bobo-Dioulasso and Banfora), Côte d'Ivoire (Abidjan and Toupah/Ousrou), Kenya (Nairobi), and Uganda (Masaka and Kampala). The protocol was approved by the local Independent Ethics Committees and/or Institutional Review Boards at each site, with central ethics approval in Burkina Faso provided by Burkina Faso Central Ethics Committee (approval number: 2017-02-023) and performed in accordance with Declaration of Helsinki and Good Clinical Practice guidelines and local regulations. The trial was registered with <u>ClinicalTrials.gov</u> NCT02564523. The study protocol, including the CON-SORT checklist, can be found in S1 Study Protocol and S1 CONSORT Checklist.

The primary objective was to assess the safety and tolerability in adolescents (12 to 17 years old) and children (4 to 11 years old) of the 2-dose heterologous vaccine regimen with Ad26. ZEBOV administered on day 1 and MVA-BN-Filo on day 29 or 57. Secondary and exploratory objectives included assessments of humoral and cellular immune responses to EBOV GP at different time points up to 1 year after Ad26.ZEBOV, and impact of baseline neutralising antibodies to Ad26 on the response.

Study participants, randomisation, and blinding

Study participants were recruited from the general population. Information was shared through community meetings, posters, and school-based conferences where volunteers and their legal guardian were invited to study sites. Signed informed consent was obtained from legal guardians, and signed informed assent was obtained from children over 6 or 12 years old (depending on country) before inclusion. On the day of randomisation, eligible participants aged 12 to 17 years or 4 to 11 years (inclusive) had to be healthy based on investigator's judgement, medical history, physical examination, vital signs, and clinical laboratory testing. Major exclusion criteria comprised any history of Ebola infection or prior exposure to EBOV (including travel to an epidemic Ebola area within 1 month of screening), previous receipt of a candidate Ebola vaccine or any experimental candidate Ad26- or MVA-based vaccine, and a known allergy or history of anaphylaxis or other serious adverse reactions to vaccines or to vaccine products.

Participants enrolled in both age cohorts were randomised 1:1 to the 28-day or 56-day dosing interval group using an interactive web response system (IWRS) provided by the sponsor, which was balanced using randomly permuted blocks and stratified by sites' peripheral blood mononuclear cell (PBMC) sampling capability. The IWRS assigned each participant a unique code, which was maintained within the IWRS and was not provided to the investigators. Participants in each group were further randomly assigned 5:1 to receive either Ad26.ZEBOV or placebo on day 1 and MVA-BN-Filo or placebo on day 29 or 57. Participants, investigators, and study staff remained blinded to the allocation of investigational products throughout the study. Vaccines and placebo were prepared by a site pharmacist who was the only unblinded member of staff. The pharmacist received the randomisation number and allocated the right study vaccine to the participant. Masking tape was used to cover the dispensing syringes containing the study vaccine/placebo allocated to each study participant.

Sample size determination

The planned sample size for the adolescent and children cohorts included 264 participants who were to receive either the 2-dose regimen or placebo, to substantially contribute to the overall safety database of the regimen. In each cohort, a total of 110 participants were to receive Ad26.ZEBOV and MVA-BN-Filo (55 participants in each 28-day or 56-day interval group); 22 participants were to receive placebo. Sample size determination was not based on formal statistical hypothesis testing. However, in case a specific adverse event (AE) was not observed, the one-sided 97.5% upper confidence limit of the true incidence rate of this AE was less than 6.5% and 3.3% for a sample size of 55 and 110 participants, respectively.

Vaccines

The heterologous 2-dose vaccine regimen comprises Ad26.ZEBOV (Zabdeno, Janssen-Cilag International N.V., Leiden, the Netherlands), a recombinant, replication-incompetent Ad26-based vector that encodes the full-length EBOV Mayinga GP, and MVA-BN-Filo (Mvabea, Bavarian Nordic, Kvistgård, Denmark), a recombinant, nonreplicating, modified vaccinia Ankara-vectored vaccine encoding EBOV Mayinga, Sudan virus Gulu, and Marburg virus Musoke variant GPs, as well as Tai Forest virus nucleoprotein. Ad26.ZEBOV containing 5×10^{10} viral particles on day 1 was followed by MVA-BN-Filo 1×10^8 infectious units (Inf.U) on day 29 or day 57. Both vaccines were supplied as frozen liquid suspensions and thawed before use. Both vaccines and placebo (0.9% saline) were administered by intramuscular injection (0.5 mL) in the deltoid.

Safety and tolerability assessments

After each vaccination, participants were assessed at 30 and 60 minutes for any immediate AEs. Solicited local and systemic AEs and daily body temperature were recorded for up to 7 days in diary cards, and unsolicited AEs were evaluated until 42 days after the second vaccination. Serious AEs (SAEs) were to be reported to investigators at any time. AEs were graded as 1 (mild), 2 (moderate), or 3 (severe) according to the adapted US Division of Microbiology and Infectious Diseases Toxicity Tables (2007) [18]. An independent data monitoring committee was established to regularly assess safety data.

Immunogenicity assessments

Blood samples were taken on day 1, day 29 or 57, day 50 or 78, day 209 or 237, and day 365. EBOV GP-specific total IgG binding antibody concentrations were measured in sera by the Filovirus Animal Nonclinical Group (FANG) ELISA at Q² Solutions (San Juan Capistrano, CA, US) [14,16,19] and were expressed as group geometric mean concentrations (GMCs) of ELISA units (EU)/mL with 95% confidence intervals (CIs). EBOV GP-specific neutralising antibody titres were measured using a pseudovirion neutralisation assay (psVNA) at Monogram (San Francisco, CA, US) [14,16] and expressed as group geometric mean titres (GMTs) of the half maximal inhibitory titre (IC₅₀) with 95% CIs.

Ad26-specific neutralising antibodies were measured using an Ad26-specific virus neutralisation assay (Ad26 VNA) at baseline (Janssen Vaccines & Prevention B.V., Leiden, the Netherlands) and were expressed as group GMTs of 90% inhibitory concentration (IC_{90}) with 95% CIs.

PBMCs, from a subset of additionally consented participants at sites capable of processing blood samples for PBMCs, were collected at the same time points specified above and were frozen for later determination of CD4+ and CD8+ T cells producing interferon (IFN)- γ ,

interleukin (IL)-2, and tumour necrosis factor (TNF)- α by intracellular cytokine staining (ICS). ICS data were expressed as the median percentage of each T cell subset (CD4+ or CD8+) producing at least one of the 3 investigated cytokines (IFN- γ , IL-2, TNF- α). IFN- γ responses to EBOV GP were evaluated using enzyme-linked immunospot (ELISpot) and were expressed as spot-forming units (SFU) per million PBMCs (reported as median reportable value). Both ICS and ELISpot were performed at the HIV Vaccine Trials Network (HVTN, Seattle, WA, USA) [11–13,20].

For sample positivity and responder definitions used in the FANG ELISA, psVNA, ELISpot, and ICS analyses, see **Text A in** <u>S1 Supporting information</u>.

Statistical analyses

The study was originally designed as a prospective study with no formal hypothesis testing. Safety is presented descriptively for the full analysis set entailing all participants who were randomised and received at least one vaccine or placebo dose irrespective of protocol deviations. Immunogenicity is presented for the per protocol set, which includes all randomised and vaccinated participants who received both vaccinations, had at least one evaluable immunogenicity sample postvaccination, and had no major protocol deviations influencing the immune response. Binding antibodies were expressed as group GMCs of EU with 95% CIs at each time point, and responder rates (i.e., the percentage of each group with postvaccination concentrations >2.5-fold the lower limit of quantification [LLOQ; 36.11 EU/mL] in baseline seronegative individuals, or >2.5-fold the baseline value in initially seropositive participants). All values below the LLOQ were imputed with half the LLOQ value.

Spearman correlation coefficients were calculated for EBOV GP-specific binding antibody concentrations (FANG ELISA) versus psVNA titres 21 days post-MVA-BN-Filo, and for FANG ELISA and psVNA (21 days post-MVA-BN-Filo) versus Ad26 VNA (baseline) data. All statistical analyses were performed by the sponsor using SAS (version 9.2, SAS Institute, Cary, NC).

While no formal statistical testing was originally planned, statistical comparisons were performed post hoc for primary and secondary outcomes. The statistical significance was set at pvalue < 0.05. No test multiplicity adjustments were performed.

Results

Participant enrolment and baseline demographics

Enrolment of participants began on February 26, 2016, and the date of last participant last visit was November 28, 2018. Following screening, 263 eligible participants were randomised and received a study vaccine: 131 adolescents and 132 children. There was good study compliance, with 125/131 (95%) adolescents and 131/132 (99%) children completing the study (Fig 1) and no vaccination-related withdrawals. Study adolescents and children had a median age of 14.0 (range 11 to 17) and 7.0 (range 4 to 11) years, respectively. Other demographics across the dosing interval groups within the 2 age cohorts were similar (Table 1).

Safety

In general, both vaccines were well tolerated in adolescents and children, with solicited AEs that were mostly mild to moderate in severity and of short duration (Fig 2; Tables A and E in <u>S1 Supporting information</u>). The frequency of AEs appeared to be unaffected by the interval between doses.



Fig 1. Study disposition of adolescents and children.

https://doi.org/10.1371/journal.pmed.1003865.g001

Table 1. Baseline characteristics of the participants; full analysis set $^{\ast}.$

		Adolescents	(12–17 years)		Children (4-11 years)					
	28-day inter	val group	56-day interval group		28-day inter	rval group	56-day interval group			
Characteristic	Active vaccines	Placebo	Active vaccines	Placebo	Active vaccines	Placebo	Active vaccines	Placebo		
N =	55	11	55	10	54	12	54	12		
Sex—No. (%)										
Male	30 (55)	6 (55)	29 (53)	6 (60)	27 (50)	7 (58)	26 (48)	7 (58)		
Female	25 (45)	5 (45)	26 (47)	4 (40)	27 (50)	5 (42)	28 (52)	5 (42)		
Black—No. (%)	55 (100)	11 (100)	55 (100)	10 (100)	54 (100)	12 (100)	54 (100)	12 (100)		
Age (years)	-									
Mean (SD)	14.4 (1.76)	14.5 (1.86)	14.1 (1.56)	14.2 (1.81)	7.6 (2.06)	7.1 (2.07)	7.8 (2.23)	7.3 (2.09)		
Median (range)	14.0 (11–17)	14.0 (12–17)	14.0 (12–17)	14.0 (12–17)	7.0 (4-11)	6.5 (4-10)	8.0 (4-11)	7.0 (4-11)		
Body mass index (kg/	⁷ m ²)									
Mean (SD)	19.36 (2.78)	18.17 (3.82)	18.95 (3.26)	18.03 (1.84)	-	-	-	-		
Weight for age percer	ntile									
Mean (SD)	-	-	-	-	29.32 (21.58)	38.69 (25.88)	28.51 (23.39)	37.91 (16.26)		

*The numbers of participants who received at least 1 dose of the study vaccines or placebo.

N, all participants who received at least 1 dose of the study vaccines or placebo; SD, standard deviation.

https://doi.org/10.1371/journal.pmed.1003865.t001

A Adverse Event	Cohort	Interval Group	Mild Injection	Moderate Severe Solicited local adverse events post-dose 1	Solicited local adverse events post-dose 2	B Adverse Event	Cohort	Interval Group	■ Mild Injection	Moderate Severe Solicited systemic adverse events post-dose 1	Solicited systemic adverse events post-dose 2
	Adolescents (12-17 years)	28-day interval group 56-day	Ad26, MVA Placebo Ad26, MVA				Adolescents (12-17 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
Any solicited local adverse events	Children (4-11 years)	group Pl 28-day Ad26 interval group P	Placebo Ad26, MVA Placebo			Any solicited systemic adverse events	Children (4-11 years)	group 28-day interval group	Placebo Ad26, MVA Placebo		
		56-day interval group	Ad26, MVA Placebo					56-day interval group	Ad26, MVA Placebo		
Erythema*	(12-17 years)	26-day interval group 56-day	Ad26, MVA Placebo Ad26, MVA			Artifraigia	(12-17 years)	zo-uay interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
	Children (4-11 years)	group 28-day interval group	Placebo Ad26, MVA Placebo			Chills	Adolescents (12-17 years)	group 28-day interval group	Placebo Ad26, MVA Placebo		
	Adolescents	56-day interval group 28-day	Ad26, MVA Placebo Ad26, MVA			Fatigue	Adolescents	56-day interval group 28-day	Ad26, MVA Placebo Ad26, MVA		
Pain	(12-17 years)	interval group 56-day interval	Placebo Ad26, MVA Placebo				(12-17 years)	interval group 56-day interval	Placebo Ad26, MVA Placebo		
	Children (4-11 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA			Headache	Adolescents (12-17 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
Pruritus	Adolescents (12-17 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA			Myalgia	Adolescents (12-17 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
Pruntus	Children (4-11 years)	group 28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA			Nausea	Adolescents (12-17 years)	group 28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
	Adolescents (12-17 years)	group 28-day interval group	Placebo Ad26, MVA Placebo				Adolescents (12-17 years) ^b	group 28-day interval group	Placebo Ad26, MVA ^c Placebo		
Swelling	Children (4-11 years)	so-day interval group 28-day interval group	Ad26, MVA Placebo Ad26, MVA Placebo			Pyrexia	Children (4-11 years) ^d	56-day interval group 28-day interval group	Ad26, MVA Placebo Ad26, MVA Placebo		
		56-day interval group	Ad26, MVA Placebo			Decreased	Children	56-day interval group	Ad26, MVA Placebo		
			(0 20 40 60 80 Percentage of participants	100 0 20 40 60 80 100 Percentage of participants	activity	(4-11 years)	interval group 56-day interval group	Ad26, MVA Placebo Ad26, MVA Placebo		
C	0 alterna	later and	Mild	Moderate Severe		Decreased appetite	Children (4-11 years)	28-day interval group 56-day interval	Ad26, MVA Placebo Ad26, MVA		
Event	Adolescents (12-17 years)	Group 28-day interval group	Ad26, MVA Placebo	events post-dose 1	events post-dose 2	Irritability	Children (4-11 years)	group 28-day interval group 56-day	Placebo Ad26, MVA Placebo Ad26, MVA		
Any unsolicited adverse events	Children (4-11 years)	56-day interval group 28-day interval	Ad26, MVA Placebo Ad26, MVA			Vomiting	Children (4-11 years)	interval group 28-day interval	Placebo Ad26, MVA		
		group 56-day interval group	Placebo Ad26, MVA Placebo					group 56-day interval group	Placebo Ad26, MVA Placebo		
			(20 40 60 80	100 0 20 40 60 80 100					Percentage of participants	Percentage of participants

Fig 2. Solicited local AEs. (A) Solicited systemic AEs (**B**) and unsolicited AEs (**C**); full analysis set. Percentages reflect n/N where n is the number of participants with one or more AEs and N is the number of participants with available reactogenicity data after the given dose (solicited AEs) or the number of participants who received the given dose (unsolicited AEs). Only unsolicited AEs reported between the dose 1 vaccination and 28 days post-dose 1, and between dose 2 vaccination and 28 days post-dose 2 are included in this table. Different diaries were used in adolescents (12–17 years) and children (4–11 years) to collect solicited systemic AEs. Solicited systemic AEs collected in adolescents: arthralgia, chills, fatigue, headache, myalgia, nausea, and pyrexia; in children: pyrexia, decreased activity, decreased appetite, irritability, and vomiting. ^aPer the DMID Toxicity Tables used in this study, erythema was graded based on the diameter data only. ^bPyrexia mild: 38.0–38.4 [°]C, moderate: 38.5–38.9 [°]C, severe: >38.9 [°]C. ^cN = 54. ^dPyrexia mild: 38.0–38.4 [°]C, moderate: 38.5–40.0 [°]C, severe: >40.0 [°]C. Ad26: Ad26.ZEBOV at a dose of 5 × 10¹⁰ viral particles; MVA: MVA-BN-Filo at a dose of 1 × 10⁸ Inf.U. Ad26, Ad26.ZEBOV; AE, adverse event; DMID, Division of Microbiology and Infectious Diseases; Inf.U, infectious units; MVA, MVA-BN-Filo.

https://doi.org/10.1371/journal.pmed.1003865.g002

One adolescent died from an unrelated typhoid fever and malaria infection with onset 52 days after MVA-BN-Filo vaccination. Two children had SAEs, a child burned in a domestic accident, and another child diagnosed with malaria, who recovered within 5 days; these were not considered by the investigator to be related to the study vaccine (**Table F in** <u>S1 Supporting information</u>).

Solicited AEs—Adolescents (12–17 years). The frequency of any solicited local AE was numerically higher in vaccinees; reported in 56/110 (51%), 49/109 (45%), 8/21 (38%), and 6/20 (30%) adolescents following Ad26.ZEBOV, MVA-BN-Filo, first and second placebo injections, respectively: p-value = 0.34 for the events following Ad26.ZEBOV compared to first placebo injection, and p-value = 0.23 for the events following MVA-BN-Filo compared to second placebo injection (**Table C in** <u>S1 Supporting information</u>). Pain at the injection site was the most frequently reported solicited local AE in adolescents. Two severe solicited local AEs were reported in adolescents (both cases of swelling): one reported in a vaccinee following MVA-BN-Filo injection.

Solicited systemic AEs were reported in 59/110 (54%), 52/109 (48%), 9/21 (43%), and 9/20 (45%) adolescents following Ad26.ZEBOV, MVA-BN-Filo, first and second placebo injections, respectively (Fig 2A and 2B; **Table A in** S1 Supporting information): *p*-value = 0.48 for the events following Ad26.ZEBOV compared to first placebo injection, and *p*-value = 1 for the events following MVA-BN-Filo compared to second placebo injection (**Table C in** S1 Supporting information). Headache was the most frequently reported solicited systemic AE in adolescents. Pyrexia was reported in identical frequencies (5%) in vaccinees (6/110 participants following Ad26.ZEBOV injection and 5/108 following MVA-BN-Filo injection) and placebo recipients (1/21 participants following first placebo injection and 1/20 following second placebo injection). Severe solicited systemic AEs were only reported by 2 adolescents (*n* = 1 fever; *n* = 1 chills), both following Ad26.ZEBOV injection.

Unsolicited AEs—Adolescents (12–17 years). The number of adolescents reporting unsolicited AE was numerically higher following injection with Ad26.ZEBOV (59/110, 54%) than MVA-BN-Filo (44/109, 40%), first placebo (9/21, 43%), or second placebo (8/20, 40%) (Fig 2C; Table A in S1 Supporting information): *p*-value = 0.48 for the events following Ad26. ZEBOV compared to first placebo injection, and *p*-value = 1 for the events following MVA-BN-Filo compared to second placebo injection (Table C in S1 Supporting information). Severe unsolicited AEs were reported in 11/110 (10%), 6/109 (6%), and 2/21 (10%) adolescents following Ad26.ZEBOV, MVA-BN-Filo and first placebo injections, respectively, and in no adolescent following second placebo injection.

Solicited AEs—**Children (4–11 years).** Solicited local AEs were reported in 55/108 (51%), 44/108 (41%), 11/24 (46%), and 4/23 (17%) children following Ad26.ZEBOV, MVA-BN-Filo, first placebo injection, and second placebo injection, respectively: *p*-value = 0.82 for the events following Ad26.ZEBOV compared to first placebo injection, and *p*-value = 0.05 for the events following MVA-BN-Filo compared to second placebo injection (**Table C in** S1 Supporting information). Compared to the adolescents, the rates were not significantly different following either Ebola vaccine (*p*-value = 1 following Ad26.ZEBOV; *p*-value = 0.6 following MVA-BN-Filo; **Table B in** S1 Supporting information). Pain at the injection site was the most frequently reported solicited local AE in children. Severe solicited local AEs were reported in 3 children; all of these occurred following Ad26.ZEBOV vaccination (severe pain [*n* = 2]; swelling [*n* = 1]).

The number of children reporting any solicited systemic AE was significantly higher following Ad26.ZEBOV (47/108, 44%) compared to MVA-BN-Filo (20/108, 19%, *p*-value < 0.001) (**Table D in** <u>S1 Supporting information</u>). The number of children reporting any solicited systemic AE was also numerically higher following Ad26.ZEBOV than first placebo (4/24, 17%) or second placebo (7/23, 30%) (Fig 2A and 2B; Table A in S1 Supporting information). The number of children reporting any solicited systemic AE was also significantly (p-value = 0.02) higher following Ad26.ZEBOV than first placebo (4/24, 17%). The number of solicited AEs following MVA-BN-Filo (20/108, 19%) was not significantly different (p-value = 0.25) compared to second placebo (7/23, 30%) (Table C in S1 Supporting information). Compared to the adolescents, the rates were not significantly different following Ad26.ZEBOV (p-value = 0.14) but were different following MVA-BN-Filo (p-value < 0.001; Table B in S1 Supporting information). Pyrexia was the most frequently reported solicited systemic AE in children following Ad26.ZEBOV (24/108, 22%) and was significantly higher than the frequency reported following MVA-BN-Filo (22% versus 4%, *p*-value < 0.001) (Table D in S1 Supporting information). Pyrexia was not reported in any children following the first placebo. Pyrexia was reported in 4/108 (4%) and 2/23 (9%) children following MVA-BN-Filo and second placebo injections (*p*-value = 0.28; **Table C in S1** Supporting information), respectively. Irritability was the most frequently reported solicited systemic AE in children following MVA-BN-Filo (12/108, 11%) and second placebo (4/23, 17%). Except for one case of severely decreased activity and appetite plus irritability, all other solicited systemic AEs were mild or moderate.

Unsolicited AEs—**Children (4–11 years).** The frequency of children reporting unsolicited AEs was similar following Ad26.ZEBOV (42/108, 39%), MVA-BN-Filo (42/108, 39%), and first placebo injection (8/24, 33%) but was higher following the second placebo injection (15/23, 65%) (Fig 2C; **Table A in S1** Supporting information): *p*-value = 0.65 for the events following Ad26.ZEBOV compared to first placebo injection, and *p*-value = 0.04 for the events following MVA-BN-Filo compared to second placebo injection (**Table C in S1** Supporting information). Compared to the adolescents, the rates were significantly different following Ad26.ZEBOV (*p*-value = 0.03) but not after MVA-BN-Filo (*p*-value = 0.89; **Table B in S1** Supporting information). One severe unsolicited AE was reported following Ad26.ZEBOV injection (neutropaenia), and two were reported following MVA-BN-Filo injection (increased transaminases [*n* = 1]; gastroenteritis [*n* = 1]).

Immunogenicity

Binding antibody responses against EBOV GP. All vaccinees displayed robust EBOV GP-specific binding antibody responses. In placebo recipients, EBOV GP-specific binding antibody levels were either low or not quantifiable at all assessed time points (Fig.3; Table G in S1 Supporting information). No relevant differences in EBOV GP-specific binding antibody responses in either adolescents or children were observed among countries (Table J in S1 Supporting information).

Binding antibody responses against EBOV GP—Adolescents (12–17 years). After Ad26. ZEBOV and prior to MVA-BN-Filo injection, 50/54 (93%) and 50/53 (94%) vaccinees showed an EBOV GP-specific binding response at day 29 and at day 57 in the 28-day and 56-day interval groups, respectively. At 21 days post-MVA-BN-Filo, 100% of vaccinees responded, with GMCs of 6,993 EU/mL (95% CI, 5,256 to 9,303) and 13,532 EU/mL (95% CI, 10,732 to 17,061) in the 28-day (n = 53) and 56-day (n = 53) interval groups, respectively (Fig 3; Table G in S1 Supporting information) (GMC ratio [95% CI] = 0.5 [0.4 to 0.7], *p*-value < 0.001; Table H in S1 Supporting information). At 6 months post-MVA-BN-Filo, responses were observed in 38/41 (93%) vaccinees in both the 28-day (GMC, 565 EU/mL [95% CI, 463 to 689]) and the 56-day (GMC, 577 EU/mL [95% CI, 454 to 734]) interval groups. Responses persisted up to day 365 in 46/50 (92%; GMC, 593 EU/mL [95% CI, 477 to 738]) and in 47/52 (90%; GMC, 541 EU/mL [95% CI, 433 to 678]) vaccinees in the 28-day and 56-day interval groups, respectively (Fig 3; Table G in S1 Supporting information).



Fig 3. GMCs of EBOV-specific binding antibodies (FANG ELISA, 95% CI) in adolescents and children. Participants administered with Ad26. ZEBOV or placebo on day 1 and MVA-BN-Filo or placebo 28 or 56 days later as indicated. Responses are expressed as GMCs (EU/mL, 95% CI). Responses in placebo groups are shown as open symbols. Grey dotted line represents the LLOQ. The points (symbols) denote GMCs, and error bars denote 95% CIs. Ad26, Ad26.ZEBOV; CI, confidence interval; EBOV, Ebola virus; EU, ELISA units; FANG ELISA, Filovirus Animal Nonclinical Group enzyme-linked immunosorbent assay; GMC, geometric mean concentration; LLOQ, lower limit of quantification; MVA, MVA-BN-Filo; Pbo, placebo.

https://doi.org/10.1371/journal.pmed.1003865.g003

Binding antibody responses against EBOV GP—Children (4–11 years). After Ad26. ZEBOV and prior to MVA-BN-Filo vaccination, 51/53 (96%) and 51/52 (98%) vaccinees showed a response at day 29 and at day 57 in the 28-day and 56-day interval groups, respectively. At 21 days post-MVA-BN-Filo, 100% of vaccinees responded, with GMCs of 8,007 EU/mL (95% CI, 6,321 to 10,142) and 17,388 EU/mL (95% CI, 12,973 to 23,306) in the 28-day (n = 53) and 56-day (n = 51) interval groups, respectively (Fig 3; Table G in S1 Supporting information) (GMC ratio [95% CI] = 0.5 [0.3 to 0.7], p-value < 0.001; Table H in S1 Supporting information). Compared to the adolescents, the GMCs of binding antibodies at 21 days

post-dose 2 were not significantly different for the children (GMC ratio [95% CI] = 1.1 [0.8 to 1.7], *p*-value = 0.47 for 28-day interval; GMC ratio [95% CI] = 1.3 [0.9 to 1.9], *p*-value = 0.18 for 56-day interval; **Table I in** S1 Supporting information). At 6 months post-MVA-BN-Filo, responses were observed in 51/52 (98%) vaccinees in both the 28-day interval group (GMC, 841 EU/mL [95% CI, 721 to 980]) and the 56-day interval group (GMC, 715 EU/mL [95% CI, 602 to 851]). Responses persisted up to day 365 in 51/53 (96%; 981 EU/mL [95% CI, 814 to 1,183]) and 51/52 (98%; 637 EU/mL [95% CI, 529 to 767]) vaccinees in the 28-day and 56-day interval groups, respectively (Fig 3; **Table G in** S1 Supporting information).

Neutralising antibody responses against EBOV GP. Most (96% to 100%) vaccinees displayed robust EBOV GP-specific neutralising antibody responses; EBOV GP-specific neutralising antibodies were not detected in any placebo recipients (**Fig A in** <u>S1 Supporting</u> <u>information</u>; **Table K in** <u>S1 Supporting information</u>). No relevant differences in EBOV GP-specific neutralising antibody responses in either adolescents or children were observed between countries (**Table L in** <u>S1 Supporting information</u>).

Neutralising antibody responses against EBOV GP—Adolescents (12–17 years). At 21 days post-MVA-BN-Filo, responses were observed in 51/53 (96%) vaccinees in the 28-day interval group and all 53/53 (100%) vaccinees in the 56-day interval group, with GMTs of 1,879 IC₅₀ titre (95% CI, 1,424 to 2,478) and 6,403 IC₅₀ titre (95% CI, 5,289 to 7,751), respectively (**Fig A and Table K in** S1 Supporting information). At day 365, responses persisted in 27/50 (54%; GMT, 251 IC₅₀ titre [95% CI, 191 to 331]) vaccinees in the 28-day interval group and 23/52 (44%; GMT, 218 IC₅₀ titre [95% CI, 174 to 273]) vaccinees in the 56-day interval group.

Neutralising antibody responses against EBOV GP—Children (4–11 years). At 21 days post-MVA-BN-Filo, responses were observed in all 53/53 (100%) vaccinees in the 28-day interval group and in 51/53 (96%) vaccinees in the 56-day interval group, with GMTs of 2,506 IC₅₀ titre (95% CI, 1,903 to 3,300) and 8,352 IC₅₀ titre (95% CI, 6,025 to 11,577), respectively (Fig A and Table K in S1 Supporting information). At day 365, responses persisted in 44/52 (85%; GMT, 447 IC₅₀ titre [95% CI, 371 to 539]) and 32/54 (59%; GMT, 275 IC₅₀ titre [95% CI, 224 to 338]) vaccinees in the 28- and 56-day interval groups, respectively.

A strong correlation between EBOV GP-specific binding antibody concentrations and neutralising antibody titres was observed across both age cohorts and vaccine dosing interval groups at 21 days post-MVA-BN-Filo (Spearman correlation coefficient: 0.660 to 0.829) (**Fig B in** <u>S1 Supporting information</u>) and at day 365 (Spearman correlation coefficient: 0.537 to 0.841) (**Fig B in** <u>S1 Supporting information</u>).

Ad26 neutralising antibodies. Prior to vaccination, most participants were positive for preexisting Ad26 neutralising antibodies (Ad26 VNA) (Table M in S1 Supporting information). In adolescents, 111/127 (87%) participants were positive (95/107 vaccinees; 16/20 placebo recipients), and GMT was 112 IC₉₀ titre (95% CI, 86 to 145). In children, 92/130 (71%) participants were positive (76/107 vaccinees; 16/23 placebo recipients), and GMT was 86 IC₉₀ titre (95% CI, 61 to 121). No correlations were observed between baseline Ad26 VNA titres and EBOV GP-specific binding antibody concentrations (FANG ELISA) at 21 days post-MVA-BN-Filo (Spearman correlation coefficients in adolescent vaccinees: -0.20 in the 28-day and -0.06 in the 56-day interval groups; in children vaccinees: 0.10 in the 28-day and 0.01 in the 56-day interval groups) (Fig C in S1 Supporting information). Similarly, no correlations were observed between preexisting Ad26-specific VNA titres and EBOV GP-specific neutralising antibody titres at 21 days post-MVA-BN-Filo (Spearman correlation coefficients in adolescent vaccinees: -0.18 in the 28-day and -0.09 in the 56-day interval groups; in children vaccinees: 0.05 in the 28-day and 0.09 in the 56-day interval groups; in children vaccinees: 0.05 in the 28-day and 0.09 in the 56-day interval groups) (Fig C in Supporting information).

Cellular immune responses against EBOV GP. No apparent differences in cellular immune responses were observed between adolescents and children, although the sample sizes were small. CD4+ and CD8+ data (Figs D and E in S1 Supporting information; Tables N and O in S1 Supporting information) and IFN- γ ELISpot data (Fig F in S1 Supporting information; Table P in S1 Supporting information) are described in Text B in S1 Supporting information.

Discussion

We have previously reported that in adults, the heterologous 2-dose Ad26.ZEBOV, MVA-BN-Filo vaccine regimen against EVD is safe, well tolerated, and immunogenic in several Phase I and II trials [11–16]. The present report confirms those observations in 12- to 17-year-old and 4- to 11-year-old participants. The vaccine regimen was well tolerated in these younger age groups, with no vaccine-related SAEs or vaccine-related dropouts. The majority of solicited local AEs and systemic AEs were mild or moderate, with few reports of severe AEs. The frequency of pyrexia after vaccine or placebo was higher in children than in adolescents, in line with a previously reported study [21].

Robust EBOV GP-specific binding and neutralising antibody levels were observed in these younger age groups, consistent with those reported in adults [11-16], with higher responses observed with a longer time interval between the 2 doses (56 versus 28 days). Although not statistically significant, greater responses in the younger age cohort were also detected, in line with our previous observations in an Ad26.ZEBOV, MVA-BN-Filo study in Sierra Leone [21,22]. Ad26 preexisting immunity was observed in the majority of participants without having an impact on the EBOV GP-specific antibody responses postvaccination, which was consistent with our previous reports for adults and children [16,21,22]. Similar antibody responder rates and levels at all time points were found across the 4 different countries. In most participants, antibodies persisted at least up to 1 year after the first vaccination; this is an encouraging result for the long-term response [23]. The relative contribution of cellular versus humoral immune responses to protection from EVD following vaccination is under debate [24–28], but both cellular and humoral immune responses were elicited by Ad26.ZEBOV, MVA-BN-Filo vaccine regimens in the current study, and T cell-mediated responses in this study are consistent with the results of 2 previous studies conducted in African adults [12,13]. Although a mechanistic correlate of protection is currently not known, studies performed in an established animal EBOV challenge model demonstrate that binding and neutralising antibodies correlate strongly with protection [29,30]. It is therefore reassuring to observe that the 2-dose heterologous regimen elicits robust humoral responses in younger age groups, and it is particularly important that this was established in these relevant populations from 4 African countries.

Although the proportion of children with EVD is usually lower compared with adults during outbreaks [31,32], infected children are more severely affected by EVD [33]. During the 2014 to 2016 EVD outbreak in West Africa, the mortality among children and adolescents younger than 15 years of age was 73% compared with the overall case fatality rate of 71% [32]. Children demonstrated shorter incubation periods and shorter intervals between symptom onset and hospitalisation or death [32]. Therefore, vaccination of children would be very relevant in an overall prophylactic vaccination strategy in countries vulnerable to future EVD outbreaks.

Study limitations include that due to PBMC shipment losses, assay failure, and/or low sample viability, fewer children and adolescents were analysed for cellular immune responses than originally planned. Also, the sample sizes were particularly small for analysis of ICS results. Another limitation is that the follow-up period was limited to 365 days, and so it was not possible to determine whether immune responses persisted beyond this time period, although modelling results are encouraging [23]. This limitation could not be avoided as a follow-up period had to be determined prior to study start. No formal statistical testing of safety or immune response data was originally planned for this study. Although post hoc statistical comparisons were performed, no direct conclusions on these comparisons between regimens or between adolescents (12 to 17 years) and children (4 to 11 years) can be made; a clinically meaningful difference in terms of binding antibody levels is not known.

In conclusion, the data from this study suggest that the same heterologous vaccine regimen, dosage, and schedule used in adults can be safely administered to children aged \geq 4 years, with acceptable reactogenicity and robust immune responses. These results have contributed to the inclusion of the paediatric population in the indication of the 2-dose heterologous regimen [8–10].

Supporting information

S1 Study Protocol.(PDF)S1 CONSORT Checklist.

(DOCX)

S1 Supporting information. Text A. Supplementary methodology. Text B. Supplementary results. Text C. EBL2002 study group (in addition to authors). Table A. Solicited local adverse events, solicited systemic adverse events, and unsolicited adverse events; full analysis set. Table B. Comparison (children [4-11 years] versus adolescents [12-17 years]) of solicited and unsolicited adverse events after each vaccination dose based on Fisher's exact test; full analysis set. Table C. Comparison (vaccinees versus placebo recipients) of solicited and unsolicited adverse events after each vaccination dose based on Fisher's exact test; full analysis set. Table D. Comparison (Ad26.ZEBOV versus MVA-BN-Filo) of solicited and unsolicited adverse events in the full analysis set based on Fisher's exact test. (Study VAC52150EBL2002; full analysis set). Table E. Duration of solicited local adverse events and solicited systemic adverse events; full analysis set. Table F. Serious adverse events, full analysis set. Table G. EBOV GP-specific binding antibody responses (ELISA units/mL): geometric mean concentrations and responder rates; per protocol set. Table H. Comparison of EBOV-GP-specific binding antibodies in adolescents [12-17 years] and children [4-11 years] in the Ebola vaccine groups; per protocol set. Table I. Comparison of EBOV-GP-specific binding antibodies in children [4-11 years] versus adolescents [12-17 years] in the Ebola vaccine groups; per protocol set. Table J. EBOV GP-specific binding antibody responses (ELISA units/mL): geometric mean concentrations and responder rates by country; per protocol analysis set. Table K. EBOV GP-specific neutralising antibody responses (psVNA; IC₅₀ titre); per protocol analysis set. Table L. EBOV GP-specific neutralising antibody responses (psVNA; IC₅₀ titre) by country; per protocol analysis set. Table M. Ad26 neutralising antibodies (Ad26 VNA; IC₉₀ titre); per protocol analysis set. Table N. EBOV GP-specific CD4+ T cell cytokine responses (ICS, % of subset); per protocol analysis set. Table O. EBOV GP-specific CD8+ T cell cytokine responses (ICS, % of subset); per protocol analysis set. Table P. EBOV GP-specific IFN-γ producing T cell responses (IFN-Y ELISpot, SFU/106 PBMC); per protocol analysis set. Fig A. EBOV GP-specific neutralising antibody responses—Regimen plot (psVNA; IC₅₀ titre); per protocol analysis set. Fig B. Spearman correlation between EBOV GP-specific binding and neutralising antibody responses 21 days post-MVA-BN-Filo; per protocol analysis set. (A) 21

days post-dose 2. (B) 364 days post-dose 1. Fig C. Correlations between Ad26-specific neutralising antibody titres at baseline and EBOV GP-specific binding and neutralising antibodies 21 days post-dose 2. (A) Anti-EBOV GP IgG ELISA at 21 days post-dose 2 by Ad26 neutralisation assay at baseline. (B) EBOV GP neutralisation assay at 21 days post-dose 2 by Ad26 neutralisation assay at baseline. Fig D. CD4+ and CD8+ T cell responses in adolescents (ICS). Fig E. CD4+ and CD8+ T cell responses in children (ICS). Fig F. EBOV GP-specific IFN- γ producing T cell responses (ELISpot). (A) Adolescents (12–17 years). (B) Children (4–11 years). (DOCX)

Acknowledgments

The authors wish to acknowledge the contribution of all participants who have taken part in the study; the partners of the EBOVAC2 consortium: the French National Institute for Health and Medical Research (Inserm), the London School of Hygiene and Tropical Medicine, the University of Oxford, and the Centre Muraz and Inserm Transfert; the members of the Scientific Advisory Board; the members of the Ethical Advisory Board; the members of the Data Safety Monitoring Board; and the research staff and all members of the Clinical Operations Group at Janssen who worked on the trial. EBL2002 study group members are listed in **Text C in** S1 Supporting information. We thank Keith Veitch (Keith Veitch Communications, Amsterdam, the Netherlands) and Chelsea McLean and Valérie Oriol-Mathieu (Janssen Vaccines & Prevention B.V., Leiden, the Netherlands) for medical writing assistance; Marialuisa Quadri (Janssen Vaccines and Prevention B.V., Leiden, the Netherlands) and Cassidy Bayley (of Ashfield MedComms) for medical writing assistance and publication coordination; and Yvonne Salzgeber (Janssen Vaccines AG, Bern, Switzerland) and Sonia Silva (Janssen Vaccines & Prevention B.V., Leiden, the Netherlands) for publication coordination.

The EBL2002 Study Group includes members of the EBOVAC 2 Executive Steering Committee (Pr Yves Levy [Inserm], Pr Peter Piot [London School of Hygiene and Tropical Medicine], and Pr Johan van Hoof [Janssen B.V.]); members of the EBOVAC 2 Clinical Steering Committee (Dr Macaya Douoguih [Janssen B.V.], Dr Cynthia Robinson [Janssen B.V.], Pr Rodolphe Thiébaut [Inserm], Dr Laura Richert [Inserm], Dr Deborah Watson-Jones [London School of Hygiene and Tropical Medicine], Pr Brian Greenwood [London School of Hygiene and Tropical Medicine], Pr Andrew J Pollard [University of Oxford], Dr Matthew D Snape [University of Oxford], Pr Nicolas Meda [Centre Muraz], Dr Houreratou Barry [Centre Muraz], Florence Chung [Inserm Transfert], and Sinéad Quigley [Inserm Transfert]); members of Janssen B.V. (Kim Offergeld [Clinical Program Leader, Global Clinical Operations], Benoit Callendret [Compound Development Team Leader], Stephanie Dincq [Clinical Project Management Lead], Camille Ferrault [Clinical Project Management Lead], Helga Pissens [Global Trial Leader], Marleen van Looveren [Regulatory Medical Writer], Sylvia van Ballaert [Regulatory Medical Writer], Tinne de Cnodder [Global Data Management Leader], and Tracy Hendrick [Clinical Programming Lead]); Clinical Working Group members (Sanne de Ridder [Clinical Trial Manager], Len Roza [ID&V Risk Management], Njinju Fogap [Lead Programmer], Rachana Gundluru [Senior Specialist Central Monitoring Manager], Olanrewaju Oladimeji [Global Data Manager], Vanessa Errijgers [Supply Management Coordinator], Maartje van Welij [Senior Independent Drug Monitor Manager], and Nicolette Muller [Clinical Research and Functional Manager]); the France EBL2002 coordination team (Pr Yves LEVY [ESC Member], Anton Ottavi, PhD [EBOVAC2 Project Coordinator] and Eugénie Destandau [Communication Officer]); members of Inserm CIC 1401, EUCLID/F-CRIN Clinical Trials Platform (Christine Schwimmer, PhD [EUCLID Executive Director], Christine Bétard [Clinical Trial Project Manager], Laetitia Moinot, PharmD [Clinical Trial Project

Manager], and Cédrick Wallet [EUCLID Operations Manager]); members of INSERM U955, Vaccine Research Institute, Université Paris-Est Créteil (Aurélie Wiedemann [Immunologist] and Christine Lacabartz [Immunologist]); Dr Albert Minga (Site Investigator) of Centre Médical de Suivi des Donneurs de Sang, Abidjan; Pr Désiré Blehou (Site Investigator) of Service de Santé au Travail SAPH, Toupah/Ousrou; members of PAC-CI, CHU Treichville, Abidjan (Dr Ida Viho, MD [Study Coordinator] and Dr Patrick Coffie [Expert]); Dr André Inwoley (Expert) of CeDReS, Université Félix Houphouët-Boigny, Abidjan; members of the KAVI Institute of Clinical Research, University of Nairobi (Nancy Thairu [Finance Manager], Dr Borna Nyaoke [Investigator], Dr Lavinia Bwisa [Investigator], Bashir Farah [Laboratory Manager], Moses Mundia [Data Manager], and Dorothy Essendi [Study Coordinator]); members of the Makerere University Walter Reed Project, Kampala (Dr Salim Wakabi [PI], Maureen Mukyala [Study Coordinator], Amir Wamala [PoR], Allan Tindikahwa [Regulatory and Compliance Manager], Dr Betty Mwesigwa [Sub-Investigator], and Ezra Musingye [Data Manager]); members of the MRC/UVRI and LSHTM Uganda Research Unit, Entebbe (Prof. Pontiano Kaleebu [Co-PI], Dr Jennifer Serwanga [Sub-Investigator/Immunologist], Dr Ggayi Abu-Baker Mustapher [Sub-Investigator/Project Coordinator], Dr Jonathan Kitonsa [Sub-Investigator/Study Clinician], Mr Paul Taire [Pharmacy Technician], Dr Laura Joan Nsangi [Sub-Investigator/Study Clinician], Mr Vincent Basajja [Community Liaisons Officer], Mr Tobias Vudriko [Safety Laboratory Technologist], Mr Ben Gombe [Immunology Laboratory Technologist], Mr Hellen Kalungi [Study Nurse], Mr Francis Kasekende [Study Pharmacist], and Dr Mary Nyantaro [Sub-Investigator/Study Clinician]); members of Centre Muraz, Bobo-Dioulasso (Dr Lionel Wilfried Ouedraogo [Laboratory Manager], Dr Armel Poda [Sub-Investigator], Dr Bachirou Tinto [Pharmacist], Dr Dramane Kania [Laboratory Manager], Dr Naalona Sandrine Hien [Sub-Investigator], Dr Guekoun Lougue [Pharmacist], Dr Innocent Valea [Coordinator], Pr Halidou Tinto [Coordinator], and Dr Ines Evelyne Da [Sub-Investigator]); members of Groupe de Recherche Action en Santé (GRAS)/Centre National de Recherche et de Formation sur le Paludisme, Ouagadougou (Dr Alfred B. Tiono [Study Coordinator], Dr Alphonse Ouedraogo [Clinical Investigator], Dr Edith Bougouma [Study Pharmacist], Dr Issa Nebie [Senior Lab Manager], Dr Diarra Amidou [Lab Manager], and Dr Daouda Ouattara [Study Physician]); and the Independent Data Monitoring Committee (Dr Bruce Mcclain, Dr Geert Molendberghs, Dr Tsiri Agbenyega, and Dr Eric Wenceslas Joseph Balayssac).

Author Contributions

- **Conceptualization:** Sodiomon B. Sirima, Laura Richert, Georgi Shukarev, Cynthia Robinson, Auguste Gaddah, Viki Bockstal, Kerstin Luhn, Rodolphe Thiébaut, Macaya Douoguih.
- **Data curation:** Zacchaeus Anywaine, Houreratou Barry, Omu Anzala, Gaudensia Mutua, Sodiomon B. Sirima, Serge Eholie, Hannah Kibuuka, Christine Bétard, Georgi Shukarev, Michael Katwere, Cynthia Robinson, Auguste Gaddah, Dirk Heerwegh, Viki Bockstal, Kerstin Luhn, Maarten Leyssen, Rodolphe Thiébaut.
- Formal analysis: Viki Bockstal, Kerstin Luhn.
- Funding acquisition: Cynthia Robinson, Rodolphe Thiébaut, Macaya Douoguih.
- **Investigation:** Zacchaeus Anywaine, Houreratou Barry, Omu Anzala, Gaudensia Mutua, Sodiomon B. Sirima, Serge Eholie, Hannah Kibuuka, Macaya Douoguih.
- Methodology: Laura Richert, Christine Lacabaratz, M. Juliana McElrath, Stephen C. De Rosa, Kristen W. Cohen, Auguste Gaddah.
- Supervision: Cynthia Robinson, Rodolphe Thiébaut, Macaya Douoguih.

Writing - original draft: Georgi Shukarev, Michael Katwere.

Writing – review & editing: Zacchaeus Anywaine, Houreratou Barry, Omu Anzala, Gaudensia Mutua, Sodiomon B. Sirima, Serge Eholie, Hannah Kibuuka, Christine Bétard, Laura Richert, Christine Lacabaratz, M. Juliana McElrath, Stephen C. De Rosa, Kristen W. Cohen, Georgi Shukarev, Michael Katwere, Cynthia Robinson, Auguste Gaddah, Dirk Heerwegh, Viki Bockstal, Kerstin Luhn, Maarten Leyssen, Rodolphe Thiébaut, Macaya Douoguih.

References

- 1. Piot P, Spencer J. From 1976 to 2018: reflections on early investigations into the Ebola virus. Trans R Soc Trop Med Hyg. 2018; 112(12):527–8. https://doi.org/10.1093/trstmh/try088 PMID: 30137538
- World Health Organization. Situation Report: Ebola Virus Disease. 2016 Jun 10. [cited 2021 Mar 25]. https://apps.who.int/iris/bitstream/handle/10665/208883/ebolasitrep_10Jun2016_eng.pdf;jsessionid= D802F9B18350E8C97D0F9FA5FEEE6ABB?sequence=1.
- 3. World Health Organization. Ebola virus disease: Democratic Republic of the Congo. External Situation Report 96. 2020 Jun 9. [cited 2021 Mar 25]. https://apps.who.int/iris/bitstream/handle/10665/332320/ SITREP_EVD_DRC_20200610-eng.pdf?sequence=1&isAllowed=y.
- Henao-Restrepo AM, Camacho A, Longini IM, Watson CH, Edmunds WJ, Egger M, et al. Efficacy and effectiveness of an rVSV-vectored vaccine in preventing Ebola virus disease: final results from the Guinea ring vaccination, open-label, cluster-randomised trial (Ebola Ça Suffit!). Lancet. 2017; 389 (10068):505–18. https://doi.org/10.1016/S0140-6736(16)32621-6 PMID: 28017403
- World Health Organization. Preliminary results on the efficacy of rVSV-ZEBOV-GP Ebola vaccine using the ring vaccination strategy in the control of an Ebola outbreak in the Democratic Republic of the Congo: an example of integration of research into epidemic response. [cited 2021 Mar 25]. https://www. who.int/csr/resources/publications/ebola/ebola-ring-vaccination-results-12-april-2019.pdf.
- US Food and Drug Administration. First FDA-approved vaccine for the prevention of Ebola virus disease, marking a critical milestone in public health preparedness and response. [cited 2021 Mar 25]. https://www.fda.gov/news-events/press-announcements/first-fda-approved-vaccine-prevention-ebolavirus-disease-marking-critical-milestone-public-health.
- 7. European Medicines Agency. Vaccine against Ebola: EMA human medicines committee grants firstever market authorisation. [cited 2021 Mar 25]. https://www.ema.europa.eu/en/news/first-vaccineprotect-against-ebola.
- European Commission. Vaccine against Ebola: European Commission grants new market authorizations. [cited 2021 Mar 25]. https://ec.europa.eu/commission/presscorner/detail/en/IP_20_1248.
- European Medicines Agency. European public assessment report: Zabdeno. [cited 2021 Mar 25]. https://www.ema.europa.eu/en/medicines/human/EPAR/zabdeno.
- European Medicines Agency. European public assessment report: Mvabea. [cited 2021 Mar 25]. https://www.ema.europa.eu/en/medicines/human/EPAR/mvabea.
- Milligan ID, Gibani MM, Sewell R, Clutterbuck EA, Campbell D, Plested E, et al. Safety and immunogenicity of novel Adenovirus type 26- and Modified Vaccinia Ankara-vectored Ebola vaccines: a randomised clinical trial. JAMA. 2016; 315(15):1610–23. https://doi.org/10.1001/jama.2016.4218 PMID: 27092831
- Mutua G, Anzala O, Luhn K, Robinson C, Bockstal V, Anumendem D, et al. Safety and immunogenicity of a 2-dose heterologous vaccine regimen with Ad26.ZEBOV and MVA-BN-Filo Ebola vaccines: 12month data from a phase 1 randomised clinical trial in Nairobi, Kenya. J Infect Dis. 2019; 220(1):57–67. https://doi.org/10.1093/infdis/jiz071 PMID: 30796816
- Anywaine Z, Whitworth H, Kaleebu P, Praygod G, Shukarev G, Manno D, et al. Safety and immunogenicity of a 2-dose heterologous vaccination regimen with Ad26.ZEBOV and MVA-BN-Filo Ebola vaccines: 12-month data from a phase 1 randomised clinical trial in Uganda and Tanzania. J Infect Dis. 2019; 220(1):46–56. https://doi.org/10.1093/infdis/jiz070 PMID: 30796818
- Pollard AJ, Launay O, Lelievre JD, Lacabaratz C, Grande S, Goldstein N, et al. Safety and immunogenicity of a two-dose heterologous Ad26.ZEBOV and MVA-BN-Filo Ebola vaccine regimen: a phase 2 randomised clinical study in Europe (EBOVAC2). 2021; 21(4):493–506. Lancet Infect Dis. <u>https://doi.org/10.1016/S1473-3099(20)30476-X</u>
- Goldstein N, Bockstal V, Bart S, Luhn K, Robinson C, Gaddah A, et al. Safety and Immunogenicity of Heterologous and Homologous 2-Dose Regimens of Adenovirus Serotype 26– and Modified Vaccinia Ankara–Vectored Ebola Vaccines: A Randomised, Controlled Phase 1 Study [published online September 16, 2020]. J Infect Dis. https://doi.org/10.1093/infdis/jiaa586 PMID: 32939546

- Barry H, Mutua G, Kibuuka H, Anywaine Z, Sirima SB, Meda N, et al. Safety and immunogenicity of 2dose heterologous Ad26.ZEBOV, MVA-BN-Filo Ebola vaccination in healthy and HIV-infected adults: A randomised, placebo-controlled Phase II clinical trial in Africa. PLoS Med. 2021; 18(10):e1003813. https://doi.org/10.1371/journal.pmed.1003813 PMID: 34714820
- World Health Organization. Weekly epidemiological record. N 22, 2021, 96, 197–216. 2021 Jun 4. [cited 2021 Sept 9]. https://apps.who.int/iris/bitstream/handle/10665/341623/WER9622-eng-fre.pdf? sequence=1&isAllowed=y
- National Institute of Allergy and Infectious Diseases. Division of Microbiology and Infectious Diseases (DMID) Adult Toxicity Table November 2007 draft. [cited 2021 Mar 25]. <u>https://www.niaid.nih.gov/sites/ default/files/dmidadulttox.pdf</u>.
- Logue J, Tuznik K, Follmann D, Grandits G, Marchand J, Reilly C, et al. Use of the Filovirus Animal Non-Clinical Group (FANG) Ebola virus immuno-assay requires fewer study participants to power a study than the Alpha Diagnostic International assay. J Virol Methods. 2018; 255:84–90. <u>https://doi.org/ 10.1016/j.jviromet.2018.02.018 PMID: 29481881</u>
- Horton H, Thomas EP, Stucky JA, Frank I, Moodie Z, Huang Y, et al. Optimization and validation of an 8color Intracellular Cytokine Staining (ICS) assay to quantify antigen-specific T Cells induced by vaccination. J Immunol Methods. 2007; 323(1):39–54. https://doi.org/10.1016/j.jim.2007.03.002 PMID: 17451739
- Afolabi MO, Ishola D, Manno D, Keshinro B, Bockstal V, Rogers B, et al. Safety and immunogenicity of the two-dose heterologous Ad26.ZEBOV and MVA-BN-Filo Ebola vaccine regimen in children in Sierra Leone: A randomised, double-blind, controlled trial. Lancet Infect Dis. 2022; 22(1):110–22. <u>https://doi.org/10.1016/S1473-3099(21)00128-6 PMID: 34529962</u>
- Ishola D, Manno D, Afolabi MO, Keshinro B, Bockstal V, Rogers B, et al. Safety and long-term immunogenicity of the two-dose heterologous Ad26.ZEBOV and MVA-BN-Filo Ebola vaccine regimen in adults in Sierra Leone: A combined open-label, non-randomised stage 1, and a randomised, double-blind, controlled stage 2 trial. Lancet Infect Dis. 2022; 22(1):97–109. <u>https://doi.org/10.1016/S1473-3099(21)</u> 00125-0 PMID: 34529963
- Pasin C, Balelli I, Van Effelterre T, Bockstal V, Solforosi L, Prague M, et al. Dynamics of the Humoral Immune Response to a Prime-Boost Ebola Vaccine: Quantification and Sources of Variation. J Virol. 2019; 93(18):e00579–19. https://doi.org/10.1128/JVI.00579-19 PMID: 31243126
- Meyer M, Malherbe DC, Bukreyev A. Can Ebola virus vaccines have universal immune correlates of protection? Trends Microbiol. 2019; 27(1):8–16. https://doi.org/10.1016/j.tim.2018.08.008 PMID: 30201511
- Corti D, Misasi J, Mulangu S, Stanley DA, Kanekiyo M, Wollen S, et al. Protective monotherapy against lethal Ebola virus infection by a potently neutralizing antibody. Science. 2016; 351(6279):1339–42. https://doi.org/10.1126/science.aad5224 PMID: 26917593
- Qiu X, Wong G, Audet J, Bello A, Fernando L, Alimonti JB, et al. Reversion of advanced Ebola virus disease in nonhuman primates with ZMapp. Nature. 2014; 514(7520):47–53. <u>https://doi.org/10.1038/nature13777 PMID</u>: 25171469
- 27. van Griensven J, Edwards T, de Lamballerie X, Semple MG, Gallian P, Baize S, et al. Evaluation of convalescent plasma for Ebola virus disease in Guinea. N Engl J Med. 2016; 374(1):33–42. <u>https://doi.org/10.1056/NEJMoa1511812 PMID: 26735992</u>
- Marzi A, Engelmann F, Feldmann F, Haberthur K, Shupert WL, Brining D, et al. Antibodies are necessary for rVSV/ZEBOV-GP-mediated protection against lethal Ebola virus challenge in nonhuman primates. Proc Natl Acad Sci U S A. 2013; 110(5):1893–8. <u>https://doi.org/10.1073/pnas.1209591110</u> PMID: 23319647
- Roozendaal R, Hendriks J, van Effelterre T, Spiessens B, Dekking L, Solforosi L, et al. Nonhuman primate to human immunobridging to infer the protective effect of an Ebola virus vaccine candidate. NPJ Vaccines. 2020; 5(1):112. https://doi.org/10.1038/s41541-020-00261-9 PMID: 33335092
- 30. Callendret B, Vellinga J, Wunderlich K, Rodriguez A, Steigerwald R, Dirmeier U, et al. A prophylactic multivalent vaccine against different filovirus species is immunogenic and provides protection from lethal infections with Ebolavirus and Marburgvirus species in non-human primates. PLoS ONE. 2018; 13(2):e0192312. https://doi.org/10.1371/journal.pone.0192312 PMID: 29462200
- WHO Ebola Response Team. Ebola virus disease in West Africa: the first 9 months of the epidemic and forward projections. N Engl J Med. 2014; 371(16):1481–95. <u>https://doi.org/10.1056/NEJMoa1411100</u> PMID: 25244186
- **32.** UNICEF Guinea. Humanitarian Situation Report 29 August 2014. [cited 2021 Mar 25]. https://www. humanitarianresponse.info/en/operations/guinea/document/unicef-guinea-humanitarian-situationreport-29-august-2014-0.
- Peacock G, Uyeki TM, Rasmussen SA. Ebola virus disease and children: what pediatric health care professionals need to know. JAMA Pediatr. 2014; 168(12):1087–8. <u>https://doi.org/10.1001/jamapediatrics.2014.2835</u> PMID: 25325785